Author’s response to reviews

Title: The Structure of Psychological Life Satisfaction: Insights from farmers and a general community sample in Australia

Authors:

Lean V O'Brien (Lean.OBrien@anu.edu.au)
Helen L Berry (Helen.Berry@canberra.edu.au)
Anthony Hogan (Anthony.Hogan@anu.edu.au)

Version: 2 Date: 8 November 2011

Author’s response to reviews:

Dear Jim,

Thank you for your feedback and for the opportunity to address your points. As directed, I have revised two points in the manuscript:

1. I have provided more detail in the Methods Section of the Abstract to give a clearer description of how different statistical methods were used to analyse the data.

2. I have included a statement in the Methods section of the manuscript (p.8) noting that the research was granted ethics approval from the Australian Bureau of Statistics’ Statistical Clearing House, which is designated as the appropriate body to give approval by Australian Government regulations.

Please let me know if anything further is needed to allow peer review.

Best wishes,

Léan O'Brien